Author’s response to reviews

Title: Prevalence of Anemia and Associated Risk Factors among Children in Northwestern Uganda: A Cross Sectional Study

Authors:

Legason Ismail (legasoni-isma@yahoo.com)

Alex Atiku (atikualex296@gmail.com)

Ronald Ssenyonga (senyoronnie@yahoo.com)

Peter Olupot-Olupot (polupotolupot@yahoo.com)

Banson Barugahare (barugahare@gmail.com)

Version: 4 Date: 30 May 2017

Author’s response to reviews:

May 30 2017

The Editor,

BMC Hematology,

236 Gray's Inn Road,

London WC1X 8HB,

United Kingdom,

Dear Sir/Madam,

Re: Point – to – point response to the review comment of BHEM-D-16-00024R3 Manuscript

We hereby submit our response to the review comment. The revised manuscript is clean – without track changes and yellow highlights. Lines and the page where the revision has been made are indicated in the detailed response here below.
Reviewer # 1:

Comment: In the new Statistical Analysis section, EPI and Epi were used. Not sure if they refer to the same thing, if so should either use EPI or Epi for abbreviation after spelling what EPI or Epi stand for.

Response:

We thank the reviewer for noticing this error. The correct expression is “Epi Info” and this has been corrected in lines 123 – 124, page 5 in the sentence “The data was entered in EPi Info software version 3.3.2 (Centre for Diseases Control and prevention, USA) and analysed using Stata version 12 (Stata Corp., Texas)”.

Yours faithfully,

BARUGAHARE JOHN BANSON, B.Sc., M.Sc., MPHIL

SENIOR LECTURER

Corresponding Author